Author's response to reviews

Title: Temporary epicardial cardiac resynchronisation versus conventional right ventricular pacing after cardiac surgery: Study protocol for a randomised control trial.

Authors:

Stuart J Russell (russellsj1@cf.ac.uk)
Christine Tan (christine.tan@wales.nhs.uk)
Peter O'Keefe (peter.o'keefe@wales.nhs.uk)
Saeed Ashraf (saeed.ashraf@wales.nhs.uk)
Afzal Zaidi (afzal.zaidi@wales.nhs.uk)
Alan G Fraser (fraserag@cf.ac.uk)
Zaheer R Yousef (zaheer.yousef@wales.nhs.uk)

Version: 6 Date: 15 February 2012

Author's response to reviews:

Dear Sir,

The title has been amended and now complies with the journal format. A section has been added to confirm the trial is still recruiting patients.